





Financial development, resource richness, eco-innovation, and sustainable development

Ahmad, Mahmood; Ahmed, Zahoor; Alvarado, Rafael; Hussain, Nazim; Khan, Sana Akbar

Published in: Journal of Environmental Management

DOI: 10.1016/j.jenvman.2023.119824

IMPORTANT NOTE: You are advised to consult the publisher's version (publisher's PDF) if you wish to cite from it. Please check the document version below.

Document Version Publisher's PDF, also known as Version of record

Publication date: 2024

Link to publication in University of Groningen/UMCG research database

Citation for published version (APA): Ahmad, M., Ahmed, Z., Alvarado, R., Hussain, N., & Khan, S. A. (2024). Financial development, resource richness, eco-innovation, and sustainable development: Does geopolitical risk matter? Journal of Environmental Management, 351, Article 119824. https://doi.org/10.1016/j.jenvman.2023.119824

Copyright Other than for strictly personal use, it is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license (like Creative Commons).

The publication may also be distributed here under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license. More information can be found on the University of Groningen website: https://www.rug.nl/library/open-access/self-archiving-pure/taverneamendment.

Take-down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from the University of Groningen/UMCG research database (Pure): http://www.rug.nl/research/portal. For technical reasons the number of authors shown on this cover page is limited to 10 maximum.

Contents lists available at ScienceDirect

Journal of Environmental Management

journal homepage: www.elsevier.com/locate/jenvman

Research article

Financial development, resource richness, eco-innovation, and sustainable development: Does geopolitical risk matter?

Mahmood Ahmad^a, Zahoor Ahmed^{b,c,d}, Rafael Alvarado^e, Nazim Hussain^{f,g,*}, Sana Akbar Khan^h

^a Business School, Shandong University of Technology, Zibo, 255000, China

^b Adnan Kassar School of Business, Lebanese American University, Beirut, 1102-2801, Lebanon

^c Department of Business Administration, Faculty of Economics, Administrative and Social Sciences, Bahçeşehir Cyprus University, Nicosia, Türkiye

^d UNEC Research Methods Application Center, Azerbaijan State University of Economics (UNEC), Istiqlaliyyat Str. 6, Baku 1001, Azerbaijan

^e Esai Business School, Universidad Espíritu Santo, Samborondon, 091650, Ecuador

^f Faculty of Economics and Business, University of Groningen, Groningen, the Netherlands

^g Faculty of Finance and Accounting, Prague University of Economics and Business, Praha, Czech Republic

h Lyon Catholic University, ESDES, 10, Place des Archives, Lyon 2, France

ARTICLE INFO

Keywords: Financial development Geopolitical risk Resource richness Green innovation Sustainable development

ABSTRACT

Financial development and geopolitical risks can significantly affect sustainable development. However, the roles of these factors in sustainable development are rarely investigated. Thus, this study takes into account the role of geopolitical risk while exploring the effects of financial development, natural resource rents, and eco-innovation on sustainable development in the Organization for Economic Co-operation and Development (OECD) countries. To this end, yearly data from 1990 to 2019 is analyzed using advanced econometric tests. The Common Correlated Effects Mean Group (CCEMG) results indicate that financial development and eco-innovation are significantly and positively related to sustainable development. Natural resource rents have a detrimental impact on sustainable development which confirms the presence of the resource curse hypothesis in OECD countries. Furthermore, the results revealed that controlling geopolitical risk is useful in fostering sustainable development. Lastly, the panel Granger causality test unveiled one-way causality from financial development, eco-innovation, natural resource rents, and geopolitical risk to sustainable development. Moreover, causalities are found from geopolitical risk to financial development, eco-innovation and natural resources. These findings suggest that OECD countries should prioritize financial development and eco-innovation policies for sustainable development while mitigating the negative effects of natural resource rents. The geopolitical risk can harm sustainable development, so policymakers should prioritie international cooperation and risk-sharing.

1. Introduction

Sustainable development has received a lot of attention in recent years due to growing awareness about the importance of balancing economic progress with environmental conservation and social wellbeing. The term "sustainable development" was introduced initially in the Brundtland Commission report in 1987, and it has since become a central objective of national and international policies. According to that report, "Sustainable development aims to meet the needs of the present generation without compromising the ability of future generations to meet their own needs" (United Nations, 1987). Sustainable development necessitates a comprehensive strategy that could incorporate social, environmental, and economic aspects into the decision-making process. In 2015, the United Nations established a worldwide structure for sustainable development known as the Sustainable Development Goals (SDGs). This framework offers a road map for reaching a better and more sustainable future (United Nations, 2015).

Financial development and sustainable development are two interconnected concepts that have received considerable interest in academic literature and policy discourse. Financial development refers to the improvement of financial systems and institutions, which enables greater access to financial services, better allocation of resources, and

E-mail addresses: mahmood@sdut.edu.cn (M. Ahmad), zahoorahmed83@yahoo.com (Z. Ahmed), rafaalvaradolopez@gmail.com (R. Alvarado), n.hussain@rug.nl (N. Hussain), skhan@univ-catholyon.fr (S.A. Khan).

https://doi.org/10.1016/j.jenvman.2023.119824

Received 29 March 2023; Received in revised form 18 November 2023; Accepted 24 November 2023 Available online 19 December 2023

0301-4797/© 2023 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).





^{*} Corresponding author. Faculty of Economics and Business, University of Groningen, Nettelbosje 2, 9747AE, Groningen, the Netherlands.

increased financial stability (Umar et al., 2020). Levine (2005) argues that the financial sector mainly contributes to economic development by affecting investment decisions, savings rates, and the pace of technological innovation. Thriving studies on the finance-growth nexus show that the financial sector is critical to economic development, despite a few opposing arguments (Boikos et al., 2022; Hung, 2023; Hunjra et al., 2021; Mtar and Belazreg, 2023). However, the research examining the relationship between financial development and sustainable development is limited. The contemporary world is confronted with formidable environmental, social, and economic predicaments. The sustainability risks are compounded by external conflicts among countries, exemplified by the ongoing Russia-Ukraine conflict and the persistent shortage of sustained financial policies, which continue to present critical challenges.

Geopolitical risk refers to the political risks that emerge due to the active engagement of nations in international affairs. Geopolitical risks encompass a range of factors, including political instability, social unrest, military conflicts, and economic challenges, among others (Caldara and Iacoviello, 2022). The multifaceted nature of geopolitical risk can impact sustainable development in several ways. Firstly, political instability and conflicts arising from geopolitical tensions can impede economic activities, infrastructure development, and foreign direct investment (Khan et al., 2023; Nguyen et al., 2022, 2023). In addition, economic unrest due to political instability can disrupt financial markets and fundamental services like healthcare, education, and green investment projects. Secondly, geopolitical risk can intensify environmental degradation and impede the effective implementation of sustainable resource management practices (Ma et al., 2023; Zhang et al., 2023). In areas that experience conflicts or political instability, there is a tendency for natural resources to be exploited in an unsustainable manner (Wang et al., 2023). Moreover, Gupta et al. (2019) argue that geopolitical risk can result in a decline in international trade which can disrupt the supply chain of goods. This, in turn, can have adverse consequences for countries that heavily rely on exports as a primary driver of their economic development. In addition, geopolitical risk can impede the flow of technology transfer, knowledge, and resources essential for eco-innovation (Lee et al., 2023). Moreover, the presence of geopolitical conflicts can shift attention and allocate resources away from cooperative international endeavors towards internal concerns, which might potentially impede the progress of eco-innovations necessary to confront urgent environmental issues. Therefore, policymakers must mitigate geopolitical risk and prioritize sustainable development. By managing geopolitical risk effectively and promoting sustainable development, countries can create more resilient and stable economies that are better equipped to navigate future challenges (Zhang et al., 2023).

Nations endowed with abundant natural resources are expected to enjoy greater prosperity and a more accelerated development trajectory than those without such valuable natural capital. Nevertheless, this inference holds true only for a few countries with abundant natural resources (Ben-Salha et al., 2021; Jović et al., 2016). Natural resources have often been viewed as a curse rather than a blessing for many nations. A substantial body of research provides evidence for the proposition that countries with limited natural resources often demonstrate higher levels of economic growth in comparison to those nations endowed with abundant resources (Hordofa et al., 2022; Shahbaz et al., 2019; Yasmeen et al., 2021). The hypothesis of the "natural resource curse," alternatively referred to as the paradox of plenty, elucidates the detrimental consequences of natural resources on economic development. Within this particular context, it is worth noting that natural resource rents can present notable obstacles to achieving sustainable development. Natural resource revenues are a financial resource for governments to allocate towards developmental initiatives. However, it is crucial to acknowledge that inadequate management of these rents can give rise to issues such as corruption, inequality, and environmental deterioration (Ahmad et al., 2020). The achievement of sustainable development necessitates the meticulous administration of natural

resource rents to guarantee their utilization in manners that yield advantages for both present and future generations. This entails the implementation of measures to prevent environmental degradation and mitigate the negative impacts on the socio-economic well-being of indigenous communities caused by resource extraction activities.

Technological innovation is critical for productivity growth, but its role in resource income growth was neglected prior to the advent of the industrial revolution. Traditional growth theories treated technology as exogenous (Solow, 1956), but endogenous growth theories recognized it as an important factor in shaping growth (Romer, 1990). The Fourth Industrial Revolution, characterized by advanced technologies, emphasizes the significance of technology in driving economic growth. However, even though technologically advanced initiatives have significantly increased global output, they have also contributed to the destruction of the natural environment, negatively impacting SDGs. The term "eco-innovation" was first introduced by Fussler and Peter (1996) as a means to address the challenge of sustainable development. Subsequently, numerous researchers and policymakers have recognized eco-innovation as a critical enabler of sustainable development. Eco-innovations can generate economic advantages, such as reduced costs, enhanced competitiveness, and efficient utilization of natural resources while fostering environmentally friendly development (Ahmad and Wu, 2022; Hao et al., 2023).

The study diligently focused on OECD countries, which consist of economically significant nations known for their well-established financial systems, strong environmental standards, and substantial impact on the world's economic policies. Although the G20, G7, and BRICS groups hold undeniable importance in the realm of international economic issues, our selection of OECD countries is primarily motivated by our study objectives. OECD countries play a pivotal role in the adoption and implementation of policies pertaining to sustainable development. By incorporating sustainability principles into their respective policies and sectors, these countries establish standards that can serve as models for others to emulate. In addition, OECD countries make significant contributions to the advancement of sustainable development globally by means of providing aid, making investments, and establishing partnerships with developing nations. These efforts facilitate economic growth that is in harmony with environmental preservation and social well-being. This study provides valuable insights into the complex relationships between financial development, resource richness, eco-innovation, sustainable development, and the influence of geopolitical risk.

Against these arguments, the aim of the study is to probe the impact of financial development (FD), natural resource rents (NRR), and ecoinnovation on sustainable development in OECD nations while considering the influence of geopolitical risk. In doing so, this study adds to the literature in a number of ways. Firstly, the study explores the effect of financial development on sustainable development. Financial development has mostly been analyzed in relation to environmental quality and GDP (Cheng et al., 2021; Lv and Li, 2021; Mtar and Belazreg, 2023), but the literature on its effects on sustainable development is scarce and contradictory. Secondly, the study investigates the influence of geopolitical risk on sustainable development. Notably, the effect of geopolitical risk on sustainable development is rarely investigated and unidentified in the context of OECD countries. Thirdly, this study assesses the effect of NRR on sustainable development. Although research on the association between NRR and sustainable development exists, earlier studies have mainly concentrated on developing and emerging countries and yielded mixed results (Ahmad et al., 2023b; Koirala and Pradhan, 2020; Lee and He, 2022). Therefore, this study endeavors to analyze this relationship in OECD countries, which are considered to be more advanced and developed. The objective is to determine whether the abundance of natural resources is a blessing or curse for these countries. Fourthly, this work explores the linkage between eco-innovation and sustainable development. Lastly, this study uses advanced econometric estimation methods robust to slope

M. Ahmad et al.

heterogeneity, cross-sectional dependence, and endogeneity.

The study is organized as follows: Section 2 presents the theoretical framework and literature review, while Section 3 details the materials and methods employed. Section 4 focuses on the presentation of results and subsequent discussion. Finally, Section 5 summarizes the study's conclusion and outlines its policy implications.

2. Literature review

2.1. Financial development and sustainable development

Financial development (FD) and sustainable development are two interconnected concepts that are crucial for the growth and prosperity of any economy. Asteriou and Spanos (2019) stressed that FD is critical for sustainable development as it provides the necessary tools to mobilize resources, manage risks, and promote inclusive growth, ultimately contributing to a more sustainable and equitable economic system. Jalilian and Kirkpatrick (2005) found that FD can enhance economic growth and reduce poverty, which is essential to sustainable development. They assert that a proficient financial system can gather and deploy savings efficiently, ultimately resulting in increased levels of economic growth and investment. Cao et al. (2021) found that FD significantly boosts green growth by funding green projects and incentivizing the adoption of green technologies. They underlined that long-term sustainable development can be achieved by a variety of strategies, including limiting the extent of financial institution expansion while confiscating governmental privileges, encouraging capital market growth, and boosting the integration of FD with technology progress. However, in another study, Cao et al. (2022) used the spatial Durbin model in China. They concluded that the development level of a financial institution in a specific province exerts a negative influence on the green growth of that province. Conversely, it has a notably positive influence on the green growth of neighboring provinces. Moreover, the size of the stock market has a significant favorable effect on green growth in various provinces.

Ahmed et al. (2022) analyzed South Asian countries' data from 2000 to 2018 and discovered that FD had a positive impact on sustainable development. Ngo et al. (2022) found a bidirectional linkage between FD and green growth in 36 countries. They further emphasized that financing strategies for sustainable infrastructure can be crucial in accomplishing SDGs within the UN 2030 agenda for sustainable development. Deng et al. (2023) probed the effect of FD on green growth in the top polluted nations, i.e., India, Russia, Japan, China, and the USA. Their outcomes indicate that the financial market's efficiency positively promotes green growth in China and Russia while it impedes green growth in USA, Japan, and India. However, the efficiency of financial institutions significantly and positively impacts green development in Japan, China, and the USA, while negatively affecting green growth in Russia and India. Yang and Ni (2022) disclosed that FD has an adverse impact on green development in Belt and Road nations.

2.2. Natural resource rents and sustainable development

The effect of NRR on sustainable development has been subject to extensive examination in both theoretical and empirical literature. Nevertheless, whether resource abundance stimulates or impedes economic growth is debatable and literature is divided on this issue. For instance, Brunnschweiler (2008) found that countries with high NRR tend to have higher growth rates in the long term. Likewise, Raheem et al. (2018) suggest that NRR contributes to lessening poverty and enhances economic development in sub-Saharan African economies. Their study argues that NRR can finance investments in human capital, infrastructure, and social services, promoting sustainable development. Huang and Zhao (2022) claim that the efficient usage and accurate management of natural resources can promote green development. However, Yu (2023) demonstrates that the overall NRR has a

detrimental effect on economic development in Afghanistan, Nepal, and Sri Lanka, adding credence to the resource curse phenomena in three nations.

Ampofo et al. (2023) gauged the effect of NRR on economic growth in the top 8 resource abundant SSA countries. Their result suggests that an expansion in NRR in the Republic of the Congo has been found to impede economic growth, providing further evidence for the resource curse phenomenon. At the same time, they fail to find a significant correlation between NRR and economic growth in the other seven countries. Fu and Liu (2023) argue that natural resource rents pose an asymmetric effect on sustainable development. They found that natural gas and mineral rents positively impact sustainable development, while forest rents impede sustainable development.

2.3. Geopolitical risk and sustainable development

Geopolitical risk has emerged as a crucial factor affecting sustainable development in recent years. Yet, few studies have delved into the association between geopolitical risk and sustainable development. Earlier studies mainly explored the linkage between geopolitical risk and sustainable development, with a focus on how geopolitical risk affects GDP and environmental sustainability. For instance, Cheng et al. (2018) found that international geopolitical risk shocks had a significant negative impact on the GDP of 38 developing economies. Akadiri et al. (2020) identified a one-way causality from geopolitical risk to economic development in Turkey. They emphasized that geopolitical risk is an essential factor that policymakers need to consider when designing strategies to foster sustainable economic growth in Turkey. However, Sweidan and Elbargathi (2022) unveiled that there is no correlation exists between economic development and geopolitical risk in the context of Saudi Arabia. However, the joint effect of oil rents and geopolitical risk on economic progress is negative in the short run and insignificant in the long term.

2.4. Eco-innovation and sustainable development

Innovative technologies are necessary to meet the increasing needs of a growing population. Mensah et al. (2019) highlighted the significance of technological advancement in ensuring the long-term sustainability of economic development. However, with the emergence of environmental and resource-related concerns, research emphasis has progressively shifted towards achieving green and sustainable development in an environmentally conscientious manner. In this context, eco-innovation has emerged as a critical driver of sustainable development. In this context, Chen et al. (2023) studied the influence of eco-innovation on green growth in the BRICS countries during 1993-2019. Their results indicate that eco-innovations and green patents positively stimulate green growth in BRICS nations. They emphasize the need for countries to promote R&D activities to facilitate the promotion of green innovation, which can aid in achieving green growth. Likewise, from the perspective of ASEAN nations, Suki et al. (2022) discovered that eco-innovation assists in reducing carbon emissions and optimizing resource utilization, implying that eco-innovation positively impacts green growth.

Mahmood et al. (2022) evaluated the effect of eco-innovation on green growth. Their outcomes revealed that eco-innovations positively contribute to green growth because green innovation contributes to the promotion of affordable and eco-friendly technologies that not only decrease environmental pollution but also provide access to modern technologies, thereby fostering sustainable economic growth. Koseoglu et al. (2022) suggested that countries' transition towards green innovation is vital for attaining sustainable development because green innovation is the only feasible way to accomplish sustainable development while preserving the environment. In addition, green innovation can also improve resource efficiency, which is a critical aspect of green growth. According to a study by Sun et al. (2023), green innovation can increase resource efficiency by improving product design and production processes, reducing waste, and promoting the circular economy, which in turn, reduces the pressure on natural resources and helps to promote sustainable growth.

2.5. Geo-political risk, financial development, natural resources, and ecoinnovation

The influence of geopolitical risk on global economic dynamics has emerged as a crucial issue, carrying significant consequences for financial markets, natural resource management, and the advancement of eco-innovation. The impact of war, conflicts, and political tensions among nations has had a deleterious influence on financial institutions and markets. For instance, Zhang and Shi (2023) found that geopolitical risk negatively affects financial development in BRICS countries. Likewise, the study of Shi and Li (2023) reveals that geopolitical risks impede financial development in China. Phan et al. (2022) posit that an escalation in geopolitical risk significantly undermines the stability of banks in the United States. Khraiche et al. (2023) concluded that geopolitical risk posed a negative impact on the stock market development in 37 countries.

Geopolitical risk profoundly influences the utilization, trade, extraction, and management of natural resources. The literature on the nexus between geopolitical risk and NRR can be categorized into three strands. The first strand of the literature suggests a negative impact of geopolitical risk on NRR (Aloui et al., 2023; Dogan et al., 2021; Li et al., 2022). Aloui et al. (2023) studied the linkage between geopolitical risk and natural resources such as coal, oil, zinc, and copper. Their findings indicate that the ongoing Russian and Ukraine conflict has a significant impact on energy markets. Dogan et al. (2021) posits that geopolitical risk decreases NRR in the developing countries. Olanipekun and Alola (2020) used the non-linear autoregressive distributed lag model (NARDL) to investigate the relationship between oil output and geopolitical risk in the Persian Gulf. Their findings indicate that upsurges in geopolitical risk are associated with a decline in oil production. The second strand of the literature posits a favorable influence of geopolitical risks on NRR. For instance, Bouoiyour et al. (2019) contend that geopolitical risk has a positive impact on resource rent. Omar et al. (2017) found that the oil market exhibits positive returns during international crises and conflicts. This finding implies that oil functions as a safe haven in periods of geopolitical instability. The favorable impact of geopolitical unpredictability on oil prices is ascribed to heightened demand for oil due to increased military, speculative, and precautionary consumption. The third strand of the literature uncovered insignificant effects of geopolitical risk on NRR (Joëts et al., 2017; Sweidan and Elbargathi, 2022).

The existing body of work examining the effects of geopolitical risk on green innovation highlights a multifaceted and intricate linkage that is shaped by a range of contextual elements. For instance, Lee et al. (2023) geopolitical risk exerts a deleterious impact on the green innovation endeavors of enterprises, primarily by magnifying the costs associated with external funding and diminishing firms' proclivity to secure capital. On the contrary, Jia et al. (2022) posit that geopolitical risk exerts a positive influence on corporate green innovation. Furthermore, the magnitude of this impact is more pronounced for state-owned enterprises, entities receiving higher government subsidies, and those engaged in overseas operations. Yang et al. (2022) propose that diminished political risk ensures consistent policy support and financial assistance for green innovation initiatives, thereby fostering sustainable development.

2.6. Literature gap

In summary, the literature review has revealed a lack of conclusive evidence concerning the impact of FD and NRR on sustainable development and limited research devoted to gauging the effects of ecoinnovation on sustainable development. Furthermore, previous studies in environmental economics have overlooked the potential impact of geopolitical risk when exploring the impact of financial development, eco-innovation, and NRR on sustainable development. In addition, many earlier studies have not effectively tackled slope heterogeneity and interdependence in panel data; thus, their findings can be biased.

3. Theoretical framework, data, and empirical methods

3.1. Theoretical framework and model construction

This study section elucidates the theoretical framework on how financial development, geopolitical risk, eco-innovation, and NRR impact sustainable development. Sustainable finance theory posits that advancing sustainable investments and financial practices can effectively leverage financial development to bolster achieving sustainable development goals. The mobilization of financial resources through financial development facilitates the availability of funding for sustainable development projects, including renewable energy generation, sustainable agriculture, and green infrastructure. Financial development can facilitate innovation by supporting research and development activities, entrepreneurial actions, and the transfer of technology (Yu et al., 2023; Zhou and Du, 2021). Financial development plays a crucial role in fostering innovation, which in turn can contribute to addressing pressing environmental issues, such as resource depletion and climate challenges. In addition, financial development can also encourage the adoption of sustainable production and consumption patterns, thereby promoting sustainable development (Elahi et al., 2022).

Geopolitical risks can tentatively hinder sustainable development by causing various detrimental effects, as posited by the theory of political economy. Geopolitical risks are predominantly associated with political elements, including regional conflicts, economic sanctions, political instability, and trade wars. These factors can substantially impact business operations through international restrictions and risk factors. One of the primary impacts of geopolitical risks on sustainable development is the decline in investment inflows, encompassing domestic and foreign investments (Esfahani, 2006). This phenomenon may arise when investors adopt a more prudent and risk-averse approach, especially in response to a perceived escalation in the likelihood of political instability or conflict within a given geographical area. Consequently, enterprises may hesitate to allocate resources towards novel initiatives or extend their activities within these circumstances, potentially impeding the progress and advancement of the economy. Geopolitical risks can restrict trade and business activities, especially due to economic sanctions or trade barriers (Nguyen et al., 2022). These limitations can potentially restrict entry into markets and access to resources, resulting in a decline in economic activity and a decrease in overall economic growth. This can ultimately undermine the accomplishment of SDGs, particularly those related to economic growth and poverty reduction (Unger and Waarden, 1999).

The ecological modernization theory supports the association between eco-innovation and sustainable development by advocating that innovation in environmental technologies can be a powerful tool for achieving sustainable development goals. This theory posits that sustainable development can be accomplished by promoting the creation and implementation of new environmentally friendly and resourceefficient technologies and processes through eco-innovation. This can help reduce environmental impacts and increase economic competitiveness, promoting social equity and improving quality of life. Additionally, Eco-innovation can help to conserve biodiversity by promoting the development of sustainable land use practices and lowering the impact of human actions on natural ecosystems.

The resource curse theory suggests that nations rich in resources (i.e., oil, gas, and minerals) have slower economic growth and development than resource-poor nations (Ross, 1999). This is because the concentration of wealth and power associated with NRR can lead to increased

corruption, unequal distribution of resources, and a lack of diversification in the economy which in turn can hinder sustainable development (Pendergast et al., 2011).

The Dutch disease is a notable consequence of NRR that significantly affects the country's effort toward sustainable development. This phenomenon arises when a nation's economy becomes excessively reliant on a specific natural resource, such as oil, gas, or minerals; it may result in the inadvertent disregard of alternative sectors, such as manufacturing or agriculture. Excessive dependence on a single sector can potentially lead to a decrease in the competitiveness of alternative industries, ultimately resulting in their contraction or potential collapse (Ben-Salha et al., 2021). An alternative viewpoint posits that the effective management of natural resource rents has the potential to foster sustainable development. According to Brunnschweiler (2008), the presence of natural resource rents in abundant resource-rich countries can facilitate the transfer of technology and foster innovation. This, in turn, can contribute to developing appropriate approaches to address environmental issues and promote sustainable production and consumption patterns. Additionally, natural resource rents can be used to finance investments in education, infrastructure, and social services, which can lead to sustainable development (Hussain et al., 2021).

In light of the theoretical underpinnings and work of Koirala and Pradhan (2020) and Hunjra et al. (2022), this study specifies the following functional form in equation (1).

$$SD_{it} = f(FD_{it}, GPR_{it}, EI_{it}, NRR_{it}, GDP_{it})$$
(1)

In equation (1), sustainable development (SD) is the function of financial development (FD), geopolitical risk (GPR), eco-innovation (EI), natural resource rents (NRR), and economic growth (GDP), respectively. The equation is formulated as expressed in equation (2) for empirical examination.

$$SD_{it} = \alpha_0 + \beta_1 F D_{it} + \beta_2 GP R_{it} + \beta_3 E I_{it} + \beta_4 NR R_{it} + \beta_5 GD P_{it} + \varepsilon_{it}$$
(2)

In equation (2), α_0 denotes the constant term while "t," and "i" refer to the time dimension and cross-sections, respectively. ε_{it} shows the error term while β_1 , β_2 , β_3 , β_4 , and β_5 represent the coefficients of FD, GPR, EI, NRR, and GDP, respectively.

3.2. Data

This study covers the yearly panel time series dataset for OECD countries from 26 OECD countries "Australia, Austria, Belgium, Canada, Czech Republic, Denmark, Finland, France, Germany, Hungary, Ireland, Italy, Japan, South Korea, Luxembourg, Netherlands, New Zealand, Norway, Poland, Portugal, Spain, Sweden, Switzerland, Turkey, the United Kingdom, and the United States" from 1990 to 2019. The duration of the study is determined due to the availability of data. The starting year of 1990 is aligned with the availability of sustainable development data. Similarly, the selection of the end year of 2019 is based on the data availability for eco-innovation.

The dependent variable (sustainable development) is quantified as adjusted net savings as a percentage of gross net income (GNI) "Adjusted net savings are equal to net national savings plus education expenditure minus energy depletion, mineral depletion, net forest depletion, and carbon dioxide and particulate emissions damage". Adjusted net savings is a commonly employed proxy to assess sustainable development in contemporary academic literature (Azam et al., 2022; Hunjra et al., 2022; Islam et al., 2022; Koirala and Pradhan, 2020). The World Bank defines adjusted net savings as "gross national savings adjusted for the annual changes in the volume of all forms of capital". As stated by Atkinson and Hamilton (2007), the evaluation of a nation's wealth encompasses the assessment of the social value of its entire capital, encompassing various forms of assets (financial, social, natural, and human). The measurement of net investment over a specific time frame serves as a means to ascertain the progression of wealth. Asset investment is also employed to assess the evolution of social welfare over a specified period. Net investment refers to the collective investment within an economy, excluding the impacts of capital stock depletion or depreciation. The measure of adjusted net savings incorporates net investment as a means of evaluating the economic sustainability within a given society, and it is particularly useful because it also covers social welfare (Hunjra et al., 2023). The data on adjusted net savings is collected from WDI (2022).

The data on the financial development index is obtained from the IMF (2022). The financial development index quantifies the depth, access, and efficiency of a country's financial institutions and financial markets, which reflect the size and scope of financial institutions, access to financial services, and the effectiveness of financial intermediation and market operations, respectively (Tao et al., 2023; Ullah et al., 2023). The geopolitical risk rating index is attained from the ICRG (2022) database. The political risk rating index stipulates a way of gauging the political stability of nations taking into account factors such as bureaucratic quality, external conflicts, internal conflicts, demographic accountability, socio-economic conditions, ethnic tensions, law & order, investment profile, religious tension, government stability, corruption, and military interference in politics. A country with a lower geopolitical risk rating indicates a higher geopolitical risk. On the other hand, if a country has a higher rating, it denotes a lower risk. The geopolitical risk rating index is commonly employed in various studies as a tool to assess and evaluate geopolitical risk ratings (Ahmad et al., 2023a; Wang et al., 2023). The data on natural resource rents (% of GDP), and economic growth (Constant, 2015\$) are collected from WDI (2022), while the data on eco-innovation (innovation in environment-related technologies) is obtained from OECD (2022). The variables' measurement, abbreviation used, and data source are provided in Table 1.

3.3. Estimation methods

3.3.1. Cross-sectional dependency test

When analyzing panel data, it is often assumed that the observations are independent and identically distributed across individuals. This assumption may not hold if there is cross-sectional dependence (CSD), which occurs when the observations of one individual in the dataset are correlated with those of another individual. The prevalence of CSD can lead to biased estimates and incorrect inferences in data analysis. Hence, conducting CSD testing before employing other panel data methodologies is imperative. For this purpose, the study used <u>Pesaran's (2004)</u> CSD test. The CD test is specified in equation (3).

$$CSD = \sqrt{\frac{2T}{N(N-1)}} \left(\sum_{i=1}^{N-1} \sum_{j=i+1}^{N} \widehat{\rho}_{ij} \right)$$
(3)

where the pairwise correlation is indicated by " $\hat{\rho}_{ij}$, N is for crosssectional units, and T denotes the study duration.

ľa	bl	e	1	

Variables measurement.		
Variable	Abb	Me

Variable	Abb.	Measurement	Data source
Sustainable development	SD	Adjusted net savings (% of GNI)	WDI (2022)
Financial	FD	This index assesses the level of	IMF
development		accessibility, efficiency, and depth of	(2022)
		financial institutes and markets.	
Geopolitical risk	GPR	Political risk rating index	ICRG
			(2022)
Eco-innovation	EI	Patents on environmental technologies (%	OECD
		of total)	(2022)
Natural resource	NRR	Natural resource rents (% of GDP)	WDI
rents			(2022)
Economic	GDP	GDP per capita (constant 2010 USD)	WDI
growth			(2022)

3.3.2. Slope homogeneity test

This study utilized Pesaran and Yamagata (2008) to assess whether the slope is homogenous or heterogenous. The examination of heterogeneity is of utmost importance as it pertains to the identification of potential variations in slope parameters among OECD countries. These variations, stemming from differences in demographic and economic structures, have the potential to impact the consistency of panel estimators. The test equations are provided in equations (4) and (5).

$$\widetilde{\Delta}_{SH} = (N)^{\frac{1}{2}} (2K)^{-\frac{1}{2}} \left(\frac{1}{N} \widetilde{S} - k \right)$$
(4)

$$\widetilde{\Delta}_{ASH} = (N)^{\frac{1}{2}} \left(\frac{2k(T-k-1)}{T+1} \right)^{-\frac{1}{2}} \left(\frac{1}{N} \widetilde{S} - k \right)$$
(5)

3.3.3. Unit root tests

CIPS and CADF are popular unit root tests used in panel data analysis. Notably, a unit root test is used to determine whether a time series is stationary or not, which is important for conducting appropriate statistical inferences. The CADF and CIPS tests proposed by Pesaran (2007) are useful for unit root analysis, as they account for CSD among the individual time series in the data. Taking into account the CSD enables the tests to offer more precise approximations of the parameters that can enhance the dependability and authenticity of the findings. The CADF test is given in equation (6).

$$\Delta CA_{i,t} = \lambda_i + \varphi_i Z_{i,t-1} + \varphi_i \overline{Z}_{t-1} + \sum_{l=0}^p \varphi_{il} \Delta \overline{CA_{t-1}} + \sum_{l=0}^p \varphi_{il} \Delta CA_{i,t-1} + \mu_{it}$$
(6)

where in equation (6), the cross-section averages are represented by \overline{CA}_{t-1} and $\Delta \overline{CA}_{t-1}$. The expression for the CIPS test is given in equation (7):

$$\widehat{CIPS} = \frac{1}{N} \sum_{i=1}^{n} CDF_i$$
⁽⁷⁾

In equation (7), the cross-sectional augmented Dickey-Fuller is denoted by CDF.

3.3.4. Panel cointegration test

The panel cointegration test is important in econometric analysis because it can help identify the underlying relationships among variables that may not be apparent in simple cross-sectional or time-series analyses. Westerlund (2007) proposed a panel cointegration test that has gained popularity in the field of econometrics. The test is centered on the bootstrap procedure and is robust to CSD, heterogeneity, and non-stationarity of the variables, which are common issues in panel data analysis. The cointegration equation is given in equation (8):

$$\alpha_{i}(L)\Delta y_{il} = \delta_{1i} + \delta_{2i}t + \alpha_{i}(y_{il-1} - \beta_{i}x_{il-1} + \lambda_{i}(L)v_{il} + e_{il}$$
(8)

In equation (8), β_i represents the vector of the cointegrating relationship between the interrelated variables y and x.

3.3.5. Long-run analysis and robustness test

After identifying the cointegration relationship, the next recommended step is to investigate the long-term cointegration coefficients of the selected variables. The Common Correlated Effects Mean Group (CCEMG) method, as proposed by Pesaran (2006), is a widely employed approach in panel data analysis for evaluating the long-term association among the variables as it effectively addresses panel data issues, such as slope heterogeneity and cross-sectional dependence (Zeqiraj et al., 2020). The test can be written as follows:

$$y_{it} = \beta_i x_{it} + \eta_i \overline{y}_{it} + \theta_i \overline{x}_{it} + \alpha_i + \gamma_i f_t + u_{it}$$
(9)

In equation (9), the symbols \overline{x}_{it} and \overline{y}_{it} represent the means of the

independent and dependent variables across different cross-sectional units, respectively. The symbol β_i represents the coefficient of the unit slope in the given context. Meanwhile, f_t , and u_{it} represent the latent common factors and the stochastic error term. Similar to the mean group (MG) technique, each panel regression is evaluated using ordinary least squares (OLS), and the MG estimators are obtained by averaging the observations within each panel unit. This test is articulated in equation (10).

$$CCEMG = \frac{1}{N} \sum_{i=1}^{N} \widehat{\beta}_i$$
(10)

In order to validate the findings derived from the CCEMG analysis, this study employs the CuP-FM and CuP-BC methodologies established by Bai et al. (2009). These methodologies provide numerous benefits, including the ability to tackle endogeneity, serial correlation, cross-sectional dependence, and heteroscedasticity. This method is expressed in equation (11):

$$\widehat{\beta}cup, \widehat{F}cup = \operatorname{armin} \frac{1}{nT^2} \sum_{i=1}^{n} (y_i - x_i\beta) M_F(y_i - x_i\beta)$$
(11)

3.3.6. Panel granger causality test

Although the results from the CCEMG provide valuable inferences, estimating the causal flow among the variables is equally imperative for sustainable growth policies. The causal directions can give policymakers and researchers valuable insights into the underlying mechanisms that drive changes in the dependent variable, allowing them to develop impactful policies and interventions. In this study, we employ the Granger causality method proposed by Dumitrescu and Hurlin (2012) to probe the causal relationships among the variables. The causality test is articulated in equation (12):

$$\delta_{i,t} = \eta_i + \sum_{k=1}^{N} \gamma_i^{(k)} \delta_{i,t-k} + \sum_{k=1}^{N} \varphi_i^{(j)} \delta_{i,t-j} + \kappa_{i,t}$$
(12)

Where in equation (12) η_i is the intercept term and is the lag order (1*K* ~ *K*). In addition, γ and φ show the autoregressive coefficients.

4. Results and discussions

Table 2 displays the Pesaran's (2004) CSD test results for the variables under investigation. The test statistics for all variables exhibit statistical significance at the 1% level, suggesting the presence of CSD among the variables. The study includes each variable's mean absolute values of pairwise correlations (ρ). These correlations range from 0.350 for SD to 0.930 for GDP, with a median value of 0.517 for EI. In general, the findings indicate that the variables within the dataset exhibit a lack of independence and demonstrate a notable level of interdependence. Consequently, it is crucial to consider and address this CSD in subsequent analyses in order to prevent biased and inefficient estimates.

Table 3 demonstrates the finding of Pesaran and Yamagata's (2008) slope heterogeneity test. The significant test values for $\tilde{\Delta}$ and $\tilde{\Delta}_{adjusted}$ demonstrate the existence of slope heterogeneity against the null hypothesis of homogeneity of slopes. These findings suggest heterogeneity

Table 2
CSD test results.

Variable	Value	P-value	Mean abs(ρ)
SD	14.621*	0.000	0.350
FD	9.453*	0.000	0.805
GPR	35.626*	0.000	0.392
EI	49.546*	0.000	0.517
NRR	31.361*	0.000	0.373
GDP	88.274*	0.000	0.930

Note. The symbol * denotes a level of significance at 1%.

Table 3

Slope heterogeneity test results.

Test	Value	Prob.
$\widetilde{\Delta}$	17.597*	0.000
$\widetilde{\Delta}_{adjusted}$	20.097*	0.000

Note. The symbol * denotes a level of significance at 1%.

in the slope parameters, and it is necessary to account for this slope heterogeneity in subsequent analyses to ensure robust and accurate outcomes.

Pesaran (2007) CADF and CIPS tests were utilized to determine the variables' stationarity under investigation, and the findings are given in Table 4. The results specify that the SD and GDP indicate a unit root issue at the level (I (0)) in the CIPS test. The findings of the CADF reveal that the SD, NRR, and GDP variables show signs of nonstationarity at the level (I (0)). Overall, the unit root test indicates that the variables' sequence of integration is inconsistent. After the first differentiation, however, all variables become stationary.

As illustrated in Table 5, the outcomes of the panel cointegration test performed by Westerlund (2007) specify the presence of cointegration among the underlying variables. The group and panel values are statistically significant at the 1% level, indicating that SD, FD, GPR, EI, NRR, and GDP move in tandem. Given that the panel dataset of OECD countries exhibits cointegration, it is suitable to obtain the long-term coefficients.

The long-run results using the CCEMG approach in Table 6 indicate that FD has a positive and statistically significant influence on sustainable development. Specifically, the coefficient of FD in the regression model is 0.551, which means that a 1% upsurge in FD is linked with a 0.551% increase in sustainable development, holding all other factors constant. This suggests that FD can play an imperative role in promoting sustainable economic growth in OECD countries. One way in which FD promotes sustainable economic growth is by increasing access to credit and financing. This allows businesses and individuals to invest in new technologies and innovations that promote sustainability and reduce the undesirable impact of economic activities on the natural environment. For instance, FD may provide loans to firms that invest in renewable energy sources like solar or wind power, reducing their dependency on fossil fuels and contributing to sustainable economic growth. The availability of credit and other financial services can help individuals and businesses invest in education, health care, and other basic needs, leading to higher incomes, improved living standards, and sustainable development. Financial development can facilitate international investment in sustainable projects and initiatives. OECD countries have the advantage of having developed financial markets and institutions, which can attract foreign investments in sustainable projects and promote sustainable development. These findings are similar to the study of Jalilian and Kirkpatrick (2005), Asteriou and Spanos (2019), and Cao et al. (2021), who also found a favorable impact of FD on sustainable development.

The geopolitical risk rating index (GPR) significantly and positively impacts sustainable development in OCED countries. As discussed

Tal	ble	4	
_	-		

Panel unit	root test.			
	CIPS		CADF	
	I(0)	I(1)	I(0)	I(1)
SD	-1.991	-5.058*	-1.960	-4.224*
FD	-2.682*	-5.389*	-2.361*	-4.071*
GPR	-2.914*	-5.316*	-2.904*	-4.455*
EI	-3.683*	-5.617*	-3.102*	-4.663*
NRR	-2.273*	-5.552*	-1.873	-4.222*
GDP	-1.857	-4.182*	-1.882	-3.086*

Note. The symbol * denotes a level of significance at 1%.

Table 5

m 11 c

ranci contegration test results.				
Statistic	Value	Z-value	P-value	Robust P-valu
Gt	-3.068*	-4.357	0.000	0.000
Ga	-9.583**	1.422	0.923	0.028
Pt	-13.212^{**}	-2.961	0.002	0.042
Pa	-8.374	-0.219	0.413	0.113

Note: The symbols * and ** denote significance levels of 1% and 5%, respectively. Bootstrapping at 600.

Table 6			
Long-run	results	(CCEMG).	

	Coefficient	Std error	Z-stat.	Prob.
FD	0.551**	0.267	2.060	0.039
GPR	0.013**	0.005	2.440	0.015
EI	0.027**	0.012	2.320	0.020
NRR	-0.161**	0.073	-2.220	0.026
GDP	1.047*	0.399	2.620	0.009
Constant	4.304*	1.439	2.990	0.003

Note: The symbols * and ** denote significance levels of 1% and 5%, respectively.

before, a country with a higher rating implies a lower geopolitical risk, whereas a lower geopolitical risk rating index indicates a high risk. Thus, statistically, a 5% improvement in the GPR can increase sustainable development by 0.013%. These results are reasonable because OECD countries are advanced economies with stable political systems compared to non-OECD countries. In addition, OECD countries have access to more resources and expertise to help them manage geopolitical risks. OECD can also benefit from diplomatic channels and international organizations like the United Nations to reduce geopolitical risks. Thus, stable political systems generally give them a greater ability to manage these risks and continue to pursue sustainable development. Geopolitical risk mitigation positively impacts sustainable development within OECD countries through various mechanisms. Firstly, reduced geopolitical tensions and conflicts significantly foster a stable and foreseeable atmosphere conducive to economic endeavors (Cao et al., 2023). The stable political and economic environment serves as a catalyst for investments, trade activities, and the expansion of businesses, thereby facilitating the process of sustainable development. Countries that experience long-term economic growth are in a more advantageous position to allocate resources toward initiatives that promote sustainable development. These initiatives may include projects focused on renewable energy, conservation of the environment, and improving social welfare.

Additionally, a reduction in geopolitical risk facilitates international cooperation and collaboration. Nations can collaborate in mutual actions to tackle worldwide issues, such as climate change, pollution, and resource depletion. Through collaboration, OECD countries can combine their resources, exchange expertise, and adopt efficient policies and strategies to attain sustainable development objectives successfully. Additionally, the mitigation of geopolitical risk has the potential to bolster stability within supply chains and improve the availability of resources. The occurrence of political instability and conflicts in specific regions has the potential to cause disruptions in supply chains, thereby resulting in economic disturbances and subsequent price escalations. By implementing strategies aimed at reducing geopolitical risks, countries can effectively guarantee consistent and dependable access to vital resources and commodities required to promote sustainable development. Finally, a decrease in geopolitical risk can positively impact investor confidence and serve as a catalyst for attracting foreign direct investment (FDI) (Nguyen et al., 2022). International investors are inclined towards long-term investments due to the appeal of political stability and diminished risk. The augmentation of FDI can infuse financial resources, specialized knowledge, and advanced technology into

initiatives aimed at sustainable development. These factors jointly contribute to pursuing sustainable development goals and creating a more sustainable future. These results are similar to the findings of Cheng et al. (2018) and Akadiri et al. (2020) but oppose the outcomes of Sweidan and Elbargathi (2022), who did not find a significant association between geopolitical risk and economic development.

The findings further revealed a positive association between ecoinnovation (EI) and sustainable development (SD). The positive coefficient suggests that for every 1% increase in eco-innovation, sustainable development increases by 0.027% in the long term. The result indicates that promoting eco-innovation is a useful strategy for attaining sustainable progress in OECD countries. One of the vital aspects of this favorable association lies in eco-innovations capacity to encounter various environmental challenges, including but not limited to climate change, pollution, and the exhaustion of natural resources. Ecoinnovation addresses these challenges through the development of cleaner and more efficient technologies, the utilization of renewable energy sources, and the implementation of sustainable production methods. Countries can facilitate the emergence and growth of green industries and sectors associated with renewable energy, waste management, and environmentally conscious products and services by allocating resources toward developing and implementing environmentally sustainable technologies and practices. The emergence of these sectors not only generates employment opportunities and fosters economic development but also serves to advance sustainable practices. The implementation of eco-innovation has the potential to enhance the overall quality of life through the provision of improved air and water quality, as well as the creation of more favorable surroundings. Moreover, eco-innovation facilitates the mitigation of social disparities and the enhancement of accessibility to sustainable solutions, assuring the equitable distribution of sustainable development advantages across diverse societal groups (Beretta, 2018). These results are consistent with Chen et al. (2023) for BRICS and Suki et al. (2022) for ASEAN countries.

The results further suggest an unfavorable impact of NRR on sustainable development in OECD nations. The coefficient for NRR is -0.161, which is statistically significant at the 5% level. This suggests that a 1% rise in NRR is associated with a 0.161% decrease in sustainable development in OECD countries. The observed negative association can be attributed to the phenomenon known as the "resource curse," which posits that countries endowed with abundant natural resources often experience lower levels of sustainable development. This implies that nations heavily dependent on revenue generated from natural resources may encounter obstacles in attaining sustainable development in the long run. Policymakers ought to take into account the importance of investing in alternative sectors, such as renewable energy, green technologies, and services, as a means to diversify the economy and diminish reliance on revenue derived from natural resources. This may help to promote sustainable development over the long term while mitigating the negative environmental and social impacts of natural resource extraction. The results of the study support the findings of Ampofo et al. (2023) but contradict the results of Brunnschweiler (2008), Raheem et al. (2018), and Huang and Zhao (2022), who argued that NRR positively contribute to sustainable development. Furthermore, the findings revealed a favorable relationship between income per capita and sustainable development in OECD nations. Specifically, the per capita income demonstrates a positive association with adjusted net savings within the OECD context. As a nation's per capita income expands, there tends to be an increased capacity for savings and investment, which can subsequently be directed towards endeavors promoting sustainable development. Economic growth facilitates the allocation of greater financial resources by individuals and the government towards sectors such as education and renewable energy, which in turn fosters sustainable development. Thus, by prioritizing sustainable investments, countries possess the ability to foster a future that encompasses sustainability and inclusivity for their current population as well as future generations.

Journal of Environmental Management 351 (2024) 119824

Table 7	
Robustness	check.

Variable	CuP-FM		CuP-BC	
	Coefficient	T-statistics	Coefficient	P-value
FD	0.644*	12.136	0.635*	16.894
GPR	0.052*	19.133	0.043*	12.620
EI	0.340*	24.810	0.122*	29.477
NRR	-0.334*	-6.352	-0.243*	-3.157
GDP	0.570*	10.233	0.452*	14.943

Note: The symbol * denotes a significance level of 1%.

BC, indicating that financial development, geopolitical risk, ecoinnovation, and economic growth have positive coefficients, while natural resource rents have a negative coefficient. Thus, the results determined from the CuP-FM and CuP-BC are reliable and similar to the outcome of CCEMG.

Table 8 presents the results of the panel Granger causality. The findings indicate that FD, EI, NRR, and GPR Granger cause SD, but not the other way around. Moreover, two-way causality exists between GDP and SD. It means policies related to GDP can Granger cause SD and vice versa. Moreover, the results indicate bidirectional causality between FD and EI. It suggests that financial development can contribute to ecoinnovation by providing firms with the necessary financial resources to invest in research and development and materials required for ecoinnovation. Furthermore, proficient financial markets offer paths for capital acquisition via initial public offerings (IPOs) or the issuance of green bonds, thereby attracting investors with a vested interest in promoting eco-innovation. Conversely, eco-innovation also Granger causes financial development. As societal awareness of environmental challenges and the preference for sustainable solutions grows, the demand for green products rises. This demand presents potential avenues for entrepreneurs and businesses to allocate resources towards research and development, thereby fostering the growth of environmentally sustainable industries and facilitating the generation of employment opportunities within these sectors. The expansion of these sectors can, in return, promote financial markets and institutions by drawing investments, facilitating the movement of capital, and encouraging the creation of dedicated financial institutions and markets that cater to the funding requirements of environmentally friendly enterprises. GPR Granger causes FD, EI, NRR, and GDP. Also, FD, EI, NRR, and GDP Granger cause GPR posing a feedback effect. These results highlighted the significance of geopolitical risk that can impact a country's FD, eco-innovation, and NRR. Geopolitical risk can harm financial development by creating instability and uncertainty, which can result in reduced investment and investor confidence, ultimately hindering FD. On the other hand, financial development can also shape geopolitical risk by promoting economic growth and stability. Geopolitical risks, such as political instability and conflicts, can disrupt the production and extraction of natural resources, reducing resource rents. It is worth noting that the existence of valuable natural resources can also serve as a catalyst for geopolitical tensions and conflicts as the competition to gain control and access to these resources becomes more intense. The reciprocal relationship between political stability and the utilization of natural resources emphasizes the significance of implementing sustainable resource management strategies and efficient conflict resolution mechanisms in order to reduce the negative impacts of geopolitical risks on resource revenues.

5. Conclusion and policy implications

The study investigates the impact of financial development, ecoinnovation, NRR, and GDP on sustainable development in OECD countries. Additionally, the study analyzes the role of geopolitical risk in this model. The study employed advanced econometric estimation methods using the data from 1990 to 2019. The empirical outcomes depict that

The robustness test in Table 7 is established on the CuP-FM and CuP- $\,$

Table 8

Panel Granger causality test.

Variable SD FD GPR EI NRR	GDP
SD – 2.767* [0.005] 1.868*** [0.061] 1.783*** [0.074] 2.351** [0.018]	3.071* [0.002]
FD 1.029 [0.303] - 2.056** [0.039] 2.641* [0.008] 3.474* [0.000]	2.790* [0.005]
GPR 0.915 [0.360] 2.611* [0.009] - 1.864*** [0.062] 2.601* [0.009]	2.975* [0.003]
EI 0.989 [0.322] 2.194** [0.028] 2.355** [0.018] - 4.531* [0.000]	7.638* [0.000]
NRR 1.069 [0.285] 0.750 [0.452] 3.192* [0.001] 1.158 [0.246] -	4.299* [0.000]
GDP 5.766* [0.000] 4.030* [0.000] 7.126* [0.000] 3.457* [0.000] 4.081* [0.000]	-

Note: The symbols *, **, and *** denote significance levels of 1%, 5%, and 10%, respectively.

financial development and eco-innovation significantly promote sustainable development in OECD countries. In addition, reducing geopolitical risk significantly and positively impacts sustainable development in these countries. However, NRR impedes sustainable development, supporting the resource curse hypothesis in OECD countries. Moreover, economic growth is significantly and positively related to sustainable development. The panel Granger causality test indicates unidirectional causality running from financial development, geopolitical risk, ecoinnovation, and NRR towards sustainable development. Moreover, geopolitical risk Granger causes financial development, eco-innovation, and NRR and vice versa.

Based on the findings of this study, the policy implications are suggested as follows: Firstly, the results suggest that there is a positive relationship between financial development and sustainable development. Therefore, it is imperative for policymakers to give precedence to initiatives that focus on bolstering financial development. These may include initiatives to enhance financial infrastructure, expand the availability of financial services, and foster financial literacy. Furthermore, it is imperative to undertake initiatives aimed at establishing a conducive atmosphere for investment, encompassing the implementation of legal and regulatory structures that safeguard the interests of investors while fostering transparency. Also, the implementation of policies that promote competition and foster innovation within the financial sector can also contribute to the advancement of sustainable development. The COP27 meeting underscored the importance of innovative climate and development funding. Nonetheless, current flows of climate financing are insufficient and are not scaling up at the requisite pace to align with the targets set forth in the Paris Agreement. The conference underscores the imperative for a global shift to a lowcarbon economy, necessitating annual investments of at least USD 4-6 trillion. To mobilize such financial resources, a rapid and comprehensive transformation of the financial system, including its structures and processes, is imperative. This transformation requires concerted efforts from governments, central banks, commercial banks, institutional investors, and other financial stakeholders (COP27, 2022).

Secondly, geopolitical risk poses a significant impact on sustainable development. The government should augment its diplomatic endeavors and conflict resolution mechanisms to mitigate geopolitical risks. Active participation in international organizations and multilateral initiatives, such as the United Nations, offers distinct advantages in promoting the attainment of amicable resolutions between nations. According to the World Economic Report (2023), escalating geopolitical conflicts, such as the Russian-Ukraine war and trade tensions between the United States and China, are pivotal to the effective functioning of the global economic system and pose a severe threat to sustainable development. Consequently, policymakers should formulate policies aimed at fortifying international cooperation and integration through trade and investment among neighboring nations to mitigate geopolitical tensions.

Thirdly, natural resource rents pose a detrimental effect on sustainable development. Therefore, it is suggested that the government should lessen reliance on natural resources by facilitating investments in alternative economic sectors, particularly the service industry and renewable energy sector. It is imperative for nations involved in resource production and importation, along with extractive industries, to collaborate in order to transition the extractive sector into a catalyst

for sustainable development characterized by low-carbon emissions, climate resilience, inclusivity, and long-term viability. The OECD Policy Dialogue on Natural Resource-based Development (PD-NR) functions as an intergovernmental forum fostering the exchange of ideas and information among OECD and non-OECD countries engaged in natural resource extraction and consumption in a sustainable manner. Fourthly, eco-innovation poses a favorable impact on sustainable development. In the present scenario, it is imperative for governments to extend financial and regulatory assistance towards eco-innovation. This entails allocating funds for research and development of environmentally friendly technologies, offering incentives to encourage businesses to adopt sustainable practices, and actively promoting the adoption of green technologies across all sectors. Furthermore, policymakers should encourage and foster collaboration among governmental entities, businesses, and academic institutions to effectively facilitate the advancement and implementation of eco-innovation. This could be accomplished by fostering public-private partnerships, establishing platforms for disseminating knowledge, and exchanging best practices. The United Nations Framework Convention on Climate Change (UNFCCC) and COP27 underscore the significance of scaling up and enhancing the effectiveness of innovation in addressing climate change and contributing to the achievement of the United Nations' Sustainable Development Goals (SDGs).

5.1. Limitations and future directions

The current study has identified a few limitations and offers recommendations for future research endeavors. One limitation of this study is its focus solely on the panel of OECD countries, which restricts the generalizability of the findings to a broader global context. Additionally, this study relies on data collected from the years 1990-2019, indicating the potential for further investigation by incorporating data from subsequent years to extend the temporal scope. Future studies can further analyze the model by employing Artificial Intelligence tools for analysis and predictions. Furthermore, the imminent COP 28 conference, scheduled at the end of the current calendar year, is poised to prioritize the attainment of reductions in global warming, aligning with the overarching goal of realizing the 1.5 °C target stipulated in the 2015 Paris Agreement, aiming to mitigate the deleterious impacts of climate change. Within this framework, prospective research endeavors may focus on the examination of parameters discussed in this paper particularly resource abundance, geopolitical risk, and eco-innovation. Given that this study does not focus on carbon neutrality targets, the objective of future inquiries can be to elucidate the influence of these factors on the realization of carbon neutrality objectives, thereby proffering nuanced policy recommendations conducive to the attainment of carbon neutrality by the mid-point of the twenty-first century.

Funding

This study is supported by the fundamental research start-up funds from Shandong University of Technology (Project No. 4033/721020).

CRediT authorship contribution statement

Mahmood Ahmad: Conceptualization, Data curation, Formal analysis, Investigation, Methodology, Writing - original draft. Zahoor Ahmed: Conceptualization, Data curation, Formal analysis, Validation, Writing - original draft. Rafael Alvarado: Writing - review & editing. Nazim Hussain: Project administration, Resources, Supervision, Writing - review & editing, Writing - review & editing.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

Data will be made available on request.

References

- Ahmad, M., Ahmed, Z., Akbar, S., Alvarado, R., 2023a. Towards environmental sustainability in E–7 countries: assessing the roles of natural resources, economic growth, country risk, and energy transition. Resour. Pol. 82, 103486 https://doi.org/ 10.1016/j.resourpol.2023.103486.
- Ahmad, M., Jiang, P., Majeed, A., Umar, M., Khan, Z., Muhammad, S., 2020. The dynamic impact of natural resources, technological innovations and economic growth on ecological footprint: An advanced panel data estimation. Resour. Policy 69, 101817. https://doi.org/10.1016/j.resourpol.2020.101817.
- Ahmad, M., Peng, T., Awan, A., Ahmed, Z., 2023b. Policy framework considering resource curse, renewable energy transition, and institutional issues: fostering sustainable development and sustainable natural resource consumption practices. Resour. Pol. 86, 104173 https://doi.org/10.1016/J.RESOURPOL.2023.104173.
- Ahmad, M., Wu, Y., 2022. Combined role of green productivity growth, economic globalization, and eco-innovation in achieving ecological sustainability for OECD economies. J. Environ. Manag. 302, 113980 https://doi.org/10.1016/J. JENVMAN.2021.113980.
- Ahmed, F., Kousar, S., Pervaiz, A., Shabbir, A., 2022. Do institutional quality and financial development affect sustainable economic growth? Evidence from South Asian countries. Borsa Istanbul Rev. 22, 189–196. https://doi.org/10.1016/j. bir.2021.03.005.
- Akadiri, S. Saint, Eluwole, K.K., Akadiri, A.C., Avci, T., 2020. Does causality between geopolitical risk, tourism and economic growth matter? Evidence from Turkey. J. Hospit. Tourism Manag. 43, 273–277. https://doi.org/10.1016/j. jhtm.2019.09.002.
- Aloui, D., Benkraiem, R., Guesmi, K., Mzoughi, H., 2023. Managing natural resource prices in a geopolitical risk environment. Resour. Pol. 83, 103628 https://doi.org/ 10.1016/j.resourpol.2023.103628.
- Ampofo, G.M.K., Laari, P.B., Ware, E.O., Shaw, W., 2023. Further investigation of the total natural resource rents and economic growth nexus in resource-abundant sub-Saharan African countries. Miner. Econ. 36, 97–121. https://doi.org/10.1007/ s13563-022-00316-4.
- Asteriou, D., Spanos, K., 2019. The relationship between financial development and economic growth during the recent crisis: evidence from the EU. Finance Res. Lett. 28, 238–245. https://doi.org/10.1016/j.frl.2018.05.011.
- Atkinson, G., Hamilton, K., 2007. Progress along the path: evolving issues in the measurement of genuine saving. Environ. Resour. Econ. 37, 43–61. https://doi.org/ 10.1007/s10640-007-9114-7.
- Azam, M., Ftiti, Z., Hunjra, A.I., Louhichi, W., Verhoeven, P., 2022. Do marketsupporting institutions promote sustainable development? Evidence from developing economies. Econ. Modell. 116, 106023 https://doi.org/10.1016/j. econmod.2022.106023.
- Bai, J., Kao, C., Ng, S., 2009. Panel cointegration with global stochastic trends. J. Econom. 149, 82–99.
- Ben-Salha, O., Dachraoui, H., Sebri, M., 2021. Natural resource rents and economic growth in the top resource-abundant countries: a PMG estimation. Resour. Pol. 74, 101229 https://doi.org/10.1016/j.resourpol.2018.07.005.
- Beretta, I., 2018. The social effects of eco-innovations in Italian smart cities. Cities 72, 115–121. https://doi.org/10.1016/j.cities.2017.07.010.
- Boikos, S., Panagiotidis, T., Voucharas, G., 2022. Financial development, reforms and growth. Econ. Modell. 108, 105734 https://doi.org/10.1016/j. econmod.2021.105734.
- Bouoiyour, J., Selmi, R., Hammoudeh, S., Wohar, M.E., 2019. What are the categories of geopolitical risks that could drive oil prices higher? Acts or threats? Energy Econ. 84, 104523 https://doi.org/10.1016/j.eneco.2019.104523.
- Brunnschweiler, C.N., 2008. Cursing the blessings? Natural resource abundance, institutions, and economic growth. World Dev. 36, 399–419. https://doi.org/ 10.1016/j.worlddev.2007.03.004.
- Caldara, D., Jacoviello, M., 2022. Measuring geopolitical risk. Am. Econ. Rev. 112, 1194–1225. https://doi.org/10.1257/aer.20191823.

- Cao, J., Law, S.H., Bin Abdul Samad, A.R., Binti W Mohamad, W.N., Wang, J., Yang, X., 2022. Effect of financial development and technological innovation on green growth—analysis based on spatial Durbin model. J. Clean. Prod. 365, 132865 https://doi.org/10.1016/j.jclepro.2022.132865.
- Cao, J., Law, S.H., Samad, A.R.B.A., Mohamad, W.N.B.W., Wang, J., Yang, X., 2021. Impact of financial development and technological innovation on the volatility of green growth—evidence from China. Environ. Sci. Pollut. Res. 28, 48053–48069. https://doi.org/10.1007/s11356-021-13828-3.
- Cao, L., Gu, M., Jin, D., Wang, C., 2023. Geopolitical risk and economic security: exploring natural resources extraction from BRICS region. Resour. Pol. 85, 103800 https://doi.org/10.1016/j.resourpol.2023.103800.
- Chen, R., Ramzan, M., Hafeez, M., Ullah, S., 2023. Green innovation-green growth nexus in BRICS: does financial globalization matter? J. Innov. Knowl. 8, 100286 https:// doi.org/10.1016/J.JIK.2022.100286.
- Cheng, C.-Y., Chien, M.-S., Lee, C.-C., 2021. ICT diffusion, financial development, and economic growth: an international cross-country analysis. Econ. Modell. 94, 662–671. https://doi.org/10.1016/j.econmod.2020.02.008.
- Cheng, C.H.J., Chiu, C.-W., Jeremy), 2018. How important are global geopolitical risks to emerging countries? Int. Econ. 156, 305–325. https://doi.org/10.1016/j. inteco.2018.05.002.
- COP27, 2022. The Sharm El-Sheikh climate implementation summit. Available at: http s://cop27.eg/assets/files/days/COP27%20INNOVATIVE%20FINANCE-DOC-01-EGY-10-22-EN.pdf.
- Deng, X., Yang, J., Ahmed, Z., Hafeez, M., Salem, S., 2023. Green growth and environmental quality in top polluted economies: the evolving role of financial institutions and markets. Environ. Sci. Pollut. Res. 30, 17888–17898. https://doi. org/10.1007/s11356-022-23421-x.
- Dogan, E., Majeed, M.T., Luni, T., 2021. Analyzing the impacts of geopolitical risk and economic uncertainty on natural resources rents. Resour. Pol. 72, 102056 https:// doi.org/10.1016/j.resourpol.2021.102056.
- Dumitrescu, E.-I.I., Hurlin, C., 2012. Testing for Granger non-causality in heterogeneous panels. Econ. Modell. 29, 1450–1460. https://doi.org/10.1016/j. econmod.2012.02.014.
- Elahi, E., Khalid, Z., Tauni, M.Z., Zhang, H., Lirong, X., 2022. Extreme weather events risk to crop-production and the adaptation of innovative management strategies to mitigate the risk: A retrospective survey of rural Punjab, Pakistan. Technovation 117, 102255. https://doi.org/10.1016/j.technovation.2021.102255.
- Esfahani, H.S., 2006. Chapter 3 A reexamination of the political economy of growth in the MENA countries. In: Nugent, J.B., Pesaran, M.H. (Eds.), Explaining Growth in the Middle East, Contributions to Economic Analysis. Elsevier, pp. 61–102. https://doi. org/10.1016/S0573-8555(06)78003-8.

Fu, R., Liu, J., 2023. Revenue sources of natural resources rents and its impact on sustainable development: evidence from global data. Resour. Pol. 80, 103226 https://doi.org/10.1016/j.resourpol.2022.103226.

Fussler, C., Peter, J., 1996. Driving Eco-Innovation: a Breakthrough Discipline for Innovation and Sustainability. Financial Times/Prentice Hall.

- Gupta, R., Gozgor, G., Kaya, H., Demir, E., 2019. Effects of geopolitical risks on trade flows: evidence from the gravity model. Eurasian Econ. Rev. 9, 515–530. https://doi. org/10.1007/s40822-018-0118-0.
- Hao, X., Li, Y., Ren, S., Wu, H., Hao, Y., 2023. The role of digitalization on green economic growth: does industrial structure optimization and green innovation matter? J. Environ. Manag. 325, 116504 https://doi.org/10.1016/j. jenvman.2022.116504.
- Hordofa, T.T., Liying, S., Mughal, N., Arif, A., Minh Vu, H., Kaur, P., 2022. Natural resources rents and economic performance: post-COVID-19 era for G7 countries. Resour. Pol. 75, 102441 https://doi.org/10.1016/j.resourpol.2021.102441.
- Huang, L., Zhao, W., 2022. The impact of green trade and green growth on natural resources. Resour. Pol. 77, 102749 https://doi.org/10.1016/j. resourcp0.2022.102749.
- Hung, N.T., 2023. Green investment, financial development, digitalization and economic sustainability in Vietnam: evidence from a quantile-on-quantile regression and wavelet coherence. Technol. Forecast. Soc. Change 186, 122185. https://doi.org/ 10.1016/j.techfore.2022.122185.
- Hunjra, A.I., Arunachalam, M., Hanif, M., 2021. Financial development-economic growth nexus: theoretical underpinnings, empirical evidence, and critical reflections. Econ. Growth Financ. Dev. 155–178 https://doi.org/10.1007/978-3-030-79003-5_9.
- Hunjra, A.I., Azam, M., Bruna, M.G., Bouri, E., 2023. A cross-regional investigation of institutional quality and sustainable development. J. Int. Financ. Mark. Inst. Money 84, 101758. https://doi.org/10.1016/j.intfin.2023.101758.
- Hunjra, A.I., Azam, M., Bruna, M.G., Verhoeven, P., Al-Faryan, M.A.S., 2022. Sustainable development: the impact of political risk, macroeconomic policy uncertainty and ethnic conflict. Int. Rev. Financ. Anal. 84, 102370 https://doi.org/10.1016/j. irfa.2022.102370.
- Hussain, M., Ye, Z., Bashir, A., Chaudhry, N.I., Zhao, Y., 2021. A nexus of natural resource rents, institutional quality, human capital, and financial development in resource-rich high-income economies. Resour. Pol. 74, 102259 https://doi.org/ 10.1016/j.resourpol.2021.102259.
- ICRG, 2022. International country risk guide. Available at: https://www.prsgroup.com/e xplore-our-products/international-country-risk-guide/. August, 2022.
- IMF, 2022. Financial development index database. Available at: https://data.imf.org/? sk=F8032E80-B36C-43B1-AC26-493C5B1CD33B.
- Islam, M.M., Ali, M.I., Ceh, B., Singh, S., Khan, M.K., Dagar, V., 2022. Renewable and non-renewable energy consumption driven sustainable development in ASEAN countries: do financial development and institutional quality matter? Environ. Sci. Pollut. Res. 29, 34231–34247. https://doi.org/10.1007/s11356-021-18488-x.

Jalilian, H., Kirkpatrick, C., 2005. Does financial development contribute to poverty reduction? J. Dev. Stud. 41, 636–656. https://doi.org/10.1080/ 00220380500002754

Jia, S., Yang, L., Zhou, F., 2022. Geopolitical risk and corporate innovation: evidence from China. J. Multinatl. Financ. Manag. 66, 100772 https://doi.org/10.1016/j. mulfin.2022.100772.

Joëts, M., Mignon, V., Razafindrabe, T., 2017. Does the volatility of commodity prices reflect macroeconomic uncertainty? Energy Econ. 68, 313–326. https://doi.org/ 10.1016/j.eneco.2017.09.017.

Jović, S., Maksimović, G., Jovović, D., 2016. Appraisal of natural resources rents and economic development. Resour. Pol. 50, 289–291. https://doi.org/10.1016/j. resourpol.2016.10.012.

Khan, K., Khurshid, A., Cifuentes-Faura, J., 2023. Is geopolitics a new risk to environmental policy in the European Union? J. Environ. Manag. 345, 118868 https://doi.org/10.1016/j.jenvman.2023.118868.

Khraiche, M., Boudreau, J.W., Chowdhury, M.S.R., 2023. Geopolitical risk and stock market development. J. Int. Financ. Mark. Inst. Money 88, 101847. https://doi.org/ 10.1016/j.intfin.2023.101847.

Koirala, B.S., Pradhan, G., 2020. Determinants of sustainable development: evidence from 12 Asian countries. Sustain. Dev. 28, 39–45. https://doi.org/10.1002/sd.1963.

Koseoglu, A., Yucel, A.G., Ulucak, R., 2022. Green innovation and ecological footprint relationship for a sustainable development: evidence from top 20 green innovator countries. Sustain. Dev. 30, 976–988. https://doi.org/10.1002/sd.2294.

Lee, C.-C., He, Z.-W., 2022. Natural resources and green economic growth: an analysis based on heterogeneous growth paths. Resour. Pol. 79, 103006 https://doi.org/ 10.1016/j.resourpol.2022.103006.

Lee, C.-C., Zhang, J., Yu, C.-H., Fang, L., 2023. How does geopolitical risk affect corporate innovation? Evidence from China's listed companies. Emerg. Mark. Finance Trade 59, 2217–2233. https://doi.org/10.1080/1540496X.2023.2170698. Levine, R., 2005. Chapter 12 finance and growth: theory and evidence. Handb. Econ.

Growth 1, 865–934. https://doi.org/10.1016/S1574-0684(05)01012-9.

Li, Y., Cong, Z., Xie, Y., Wang, Y., Wang, H., 2022. The relationship between green finance, economic factors, geopolitical risk and natural resources commodity prices: evidence from five most natural resources holding countries. Resour. Pol. 78, 102733 https://doi.org/10.1016/j.resourpol.2022.102733.

Lv, Z., Li, S., 2021. How financial development affects CO2 emissions: a spatial econometric analysis. J. Environ. Manag. 277, 111397 https://doi.org/10.1016/j. jenvman.2020.111397.

Ma, X., Yu, T., Jiang, Q., 2023. Does geopolitical risk matter in carbon and crude oil markets from a multi-timescale perspective? J. Environ. Manag. 346, 119021 https://doi.org/10.1016/j.jenvman.2023.119021.

Mahmood, N., Zhao, Y., Lou, Q., Geng, J., 2022. Role of environmental regulations and eco-innovation in energy structure transition for green growth: evidence from OECD. Technol. Forecast. Soc. Change 183, 121890. https://doi.org/10.1016/j. techfore.2022.121890.

- Mensah, C.N., Long, X., Dauda, L., Boamah, K.B., Salman, M., Appiah-Twum, F., Tachie, A.K., 2019. Technological innovation and green growth in the organization for economic cooperation and development economies. J. Clean. Prod. 240, 118204 https://doi.org/10.1016/j.jclepro.2019.118204.
 Mtar, K., Belazreg, W., 2023. On the nexus of innovation, trade openness, financial
- Mtar, K., Belazreg, W., 2023. On the nexus of innovation, trade openness, financial development and economic growth in European countries: new perspective from a GMM panel VAR approach. Int. J. Financ. \& Econ. 28, 766–791. https://doi.org/ 10.1002/iife.2449.
- Ngo, T., Trinh, H.H., Haouas, I., Ullah, S., 2022. Examining the bidirectional nexus between financial development and green growth: international evidence through the roles of human capital and education expenditure. Resour. Policy 79, 102964. https://doi.org/10.1016/j.resourpol.2022.102964.
- Nguyen, D.T., Le, T.H., Do, D.D., Nguyen, H.N., 2023. Does geopolitical risk hinder sustainable development goals? Evidence from a panel analysis. J. Environ. Manage. 347, 119204 https://doi.org/10.1016/j.jenvman.2023.119204.
- Nguyen, T.T.T., Pham, B.T., Sala, H., 2022. Being an emerging economy: to what extent do geopolitical risks hamper technology and FDI inflows? Econ. Anal. Policy 74, 728–746. https://doi.org/10.1016/j.eap.2022.04.005.

OECD, 2022. OECD data available at: https://data.oecd.org/.

Olanipekun, I.O., Alola, A.A., 2020. Crude oil production in the Persian Gulf amidst geopolitical risk, cost of damage and resources rents: is there asymmetric inference? Resour. Policy 69, 101873. https://doi.org/10.1016/j.resourpol.2020.101873.

Omar, A.M.A., Wisniewski, T.P., Nolte, S., 2017. Diversifying away the risk of war and cross-border political crisis. Energy Econ 64, 494–510. https://doi.org/10.1016/j. eneco.2016.02.015.

Pendergast, S.M., Clarke, J.A., Van Kooten, G.C., 2011. Corruption, development and the curse of natural resources. Can. J. Polit. Sci. Can. Sci. Polit. 44, 411–437. https://doi. org/10.1017/S0008423911000114.

Pesaran, H., Yamagata, T., 2008. Testing slope homogeneity in large panels. J. Econom. 142, 50–93. https://doi.org/10.1016/j.jeconom.2007.05.010.

Pesaran, M.H., 2007. A simple panel unit root test in the presence of cross section dependence. J. Appl. Econom. 47, 265–312. https://doi.org/10.1002/jae.

Pesaran, M.H., 2006. Estimation and inference in large heterogeneous panels with a multifactor error structure. Econometrica 74, 967–1012. https://doi.org/10.1111/ j.1468-0262.2006.00692.x.

Pesaran, M.H., 2004. General Diagnostic Tests for Cross-Sectional Dependence in Panels. Univ, Cambridge, Cambridge Work, p. 435. https://doi.org/10.17863/CAM.5113. Pap. Econ.

Phan, D.H.B., Tran, V.T., Iyke, B.N., 2022. Geopolitical risk and bank stability. Financ. Res. Lett. 46, 102453 https://doi.org/10.1016/j.frl.2021.102453. Raheem, I.D., Isah, K.O., Adedeji, A.A., 2018. Inclusive growth, human capital development and natural resource rent in SSA. Econ. Chang. Restruct. 51, 29–48. https://doi.org/10.1007/s10644-016-9193-y.

- Romer, P.M., 1990. Endogenous technological change. J. Polit. Econ 98. https://doi.org/ 10.1086/261725.
- Ross, M.L., 1999. The political economy of the resource curse. World Polit 51, 297–322. https://doi.org/10.1017/S0043887100008200.
- Shahbaz, M., Ahmed, K., Tiwari, A.K., Jiao, Z., 2019. Resource curse hypothesis and role of oil prices in USA. Resour. Policy 64, 101514. https://doi.org/10.1016/j. resourpol.2019.101514.

Shi, Z., Li, Y., 2023. China's sustainable development perspective of financial development from the lens of geopolitical risk and resources extraction. Resour. Policy 86, 104298. https://doi.org/10.1016/j.resourpol.2023.104298.

Solow, R.M., 1956. A contribution to the theory of economic growth. Q. J. Econ. 70, 65–94. https://doi.org/10.2307/1884513.

Suki, N.M., Mohd Suki, N., Afshan, S., Sharif, A., Ariff Kasim, M., Rosmaini Mohd Hanafi, S., 2022. How does green technology innovation affect green growth in ASEAN-6 countries? Evidence from advance panel estimations. Gondwana Res 111, 165–173. https://doi.org/10.1016/j.gr.2022.06.019.

Sun, Y., Gao, P., Tian, W., Guan, W., 2023. Green innovation for resource efficiency and sustainability: empirical analysis and policy. Resour. Policy 81, 103369. https://doi. org/10.1016/j.resourpol.2023.103369.

Sweidan, O.D., Elbargathi, K., 2022. The effect of oil rent on economic development in Saudi Arabia: comparing the role of globalization and the international geopolitical risk. Resour. Policy 75, 102469. https://doi.org/10.1016/j.resourpol.2021.102469.

Tao, M., Sheng, M.S., Wen, L., 2023. How does financial development influence carbon emission intensity in the OECD countries: some insights from the information and communication technology perspective. J. Environ. Manage. 335, 117553 https:// doi.org/10.1016/j.jenvman.2023.117553.

Ullah, S., Adebayo, T.S., Irfan, M., Abbas, S., 2023. Environmental quality and energy transition prospects for G-7 economies: the prominence of environment-related ICT innovations, financial and human development. J. Environ. Manage. 342, 118120 https://doi.org/10.1016/j.jenvman.2023.118120.

Umar, M., Ji, X., Kirikkaleli, D., Xu, Q., 2020. COP21 Roadmap: do innovation, financial development, and transportation infrastructure matter for environmental sustainability in China? J. Environ. Manage. 271, 111026 https://doi.org/10.1016/j. ienvman.2020.111026.

Unger, B., Waarden, F. Van, 1999. Interest associations and economic growth: a critique of Mancur Olson's rise and decline of nations. Rev. Int. Polit. Econ. 6, 425–467. https://doi.org/10.1080/096922999347128.

United Nations, 2015. Sustainable development goals. United nations department of economic and social affairs, New York, United States. Available at: https://sdgs.un.org/.

United Nations, 1987. Report of the world commission on environment and development (WCED): Our Common Future. Brundtland Report. https://www.are.admin.ch/a re/en/home/media/publications/sustainable-development/brundtland-report.html.

Wang, Z., Y, C., Zhang, B., Ahmed, Z., Ahmad, M., 2023. Environmental degradation, renewable energy, and economic growth nexus: assessing the role of financial and political risks? J. Environ. Manage. 325, 116678 https://doi.org/10.1016/j. ienvman.2022.116678.

WDI, 2022. World devleopment indicators (WDI). Available at: https://datatopics.worl dbank.org/world-development-indicators/.

Westerlund, J., 2007. Testing for error correction in panel data. Oxf. Bull. Econ. Stat. 69, 709–748. https://doi.org/10.1111/j.1468-0084.2007.00477.x.

World Economic Report, 2023. The Global Risks Report 2023, eighteenth ed. Available at: https://www3.weforum.org/docs/WEF_Global_Risks_Report_2023.pdf.

Yang, H.-C., Cai, Y.-F., Zhang, M.-Y., 2022. Political risk and green technology improvement: new insights from global evidence. Innov. Green Dev. 1, 100004 https://doi.org/10.1016/j.igd.2022.100004.

Yang, L., Ni, M., 2022. Is financial development beneficial to improve the efficiency of green development? Evidence from the "Belt and Road" countries. Energy Econ 105, 105734. https://doi.org/10.1016/j.eneco.2021.105734.

Yasmeen, H., Tan, Q., Zameer, H., Vo, X.V., Shahbaz, M., 2021. Discovering the relationship between natural resources, energy consumption, gross capital formation with economic growth: can lower financial openness change the curse into blessing. Resour. Policy 71, 102013. https://doi.org/10.1016/j.resourpol.2021.102013.

Yu, H., Wang, J., Hou, J., Yu, B., Pan, Y., 2023. The effect of economic growth pressure on green technology innovation: do environmental regulation, government support, and financial development matter? J. Environ. Manage. 330, 117172 https://doi. org/10.1016/j.jenvman.2022.117172.

Yu, Y., 2023. Role of Natural resources rent on economic growth: fresh empirical insight from selected developing economies. Resour. Policy 81, 103326. https://doi.org/ 10.1016/j.resourpol.2023.103326.

Zeqiraj, V., Sohag, K., Soytas, U., 2020. Stock market development and low-carbon economy: the role of innovation and renewable energy. Energy Econ 91, 104908. https://doi.org/10.1016/j.eneco.2020.104908.

Zhang, D., Chen, X.H., Lau, C.K.M., Cai, Y., 2023. The causal relationship between green finance and geopolitical risk: implications for environmental management.

J. Environ. Manage. 327, 116949 https://doi.org/10.1016/j.jenvman.2022.116949.Zhang, J., Shi, S., 2023. Extraction of natural resources and geopolitical risk revisited: a novel perspective of research and development with financial development. Resour. Policy 85, 103799. https://doi.org/10.1016/j.resourpol.2023.103799.

Zhou, X., Du, J., 2021. Does environmental regulation induce improved financial development for green technological innovation in China? J. Environ. Manage. 300, 113685 https://doi.org/10.1016/j.jenvman.2021.113685.